

April 2, 2017

Fang-Fang Ji,  
Science Editor, Editorial Office  
Baishideng Publishing Group Inc  
E-mail: f.f.ji@wjgnet.com

Re: Manuscript NO.: 33313

Dear Dr. Ji,

Thank you for reviewing our manuscript and offering constructive criticism for improvement. We have made changes and they are summarized below.

1. Reviewer's code: 00068420:
  - a. We modified the title of the article to better reflect the content and purpose of this Minireview.
    - i. Original title: Impact of Direct Acting Antiviral Agents on Kidney Function in Hepatitis C Virus Infected Patients.
    - ii. Suggested new title: Opportunities for Treatment of the Hepatitis C Virus-Infected Patient with Chronic Kidney Disease.
2. Reviewer's code: 00054275:
  - a. Misprint errors were corrected.
3. Reviewer's codes: 01221188, 00070577, 00069297:
  - a. No comments from the reviewers were made.
4. Editor's suggestions in the edited manuscript file:
  - a. Changes were made to the manuscript according to the Guidelines and Requirements for Manuscript Revision-Minireview.
  - b. The power point version of the tables and graphs was provided to allow editing.

Once again thank you for considering our manuscript for publication.

Sincerely yours,

Marco Ladino M.D., Fernando Pedraza M.D., David Roth M.D.